Author's response to reviews

Title: Detection of Drug Effects on Gastric Emptying and Contractility Using a Wireless Motility Capsule

Authors:

Inna Rosov-Ung (inna.rozov.ung@gmail.com)
Amjad Mreyoud (ameyroud@hotmail.com)
John Moore (jmoor@fdc.org)
Gregory E Wilding (gwilding@buffalo.edu)
Elias Khawam (ekhawam@gmail.com)
Jeffrey M Lackner (lackner@buffalo.edu)
John R Semler (Jack.Semler@givenimaging.com)
Michael D Sitrin (mdsitrin@buffalo.edu)

Version: 3 Date: 17 December 2013

Author's response to reviews:

December 13, 2013

Biomedical Central

In response to the editor’s comment we have made the following revision in the manuscript:

1. A statement has been added in the Methods section stating that written informed consent for participation in the study was obtained from participants.

We thank the editor for this comment, and hope that the revised manuscript is now acceptable for publication.

Sincerely,

Michael D. Sitrin, MD
Professor of Medicine
Division of Gastroenterology, Hepatology, and Nutrition
University at Buffalo, State University of New York
Fellowship program director
Tel: 716-898-5487
Fax: 716-898-5876
Email: mdsitrin@buffalo.edu